

Key Information				
Author:	(Alan Underdown, Manager of Borough Operations,			
	Environmental Services)			
Accountable Director:	(James Hendry, Assistant Director Environmental			
	Services)			
Date of Cabinet Approval:	13/03/2025			
Service Impact Assessment (SIA)	To be reviewed and published.			
Consultee(s)				
Groups and/or individuals consulted	Borough wide consultations on collection models:			
during the development of this	2021: Over 7000 responses			
Policy:	2022: Over 5500 responses			
	2023: Over 4000 responses			
	2024: Over 7222 responses			
	We have engaged with community groups on			
	service changes:			
	<ul> <li>Community and Diversity Council</li> </ul>			
	- Women's Institute			
	<ul> <li>Disability Partnership Board</li> </ul>			
	- Sheltered Schemes			
	Regular roadshows are held across the Borough			
Groups and/or individuals consulted	Waste Strategy Working Group			
on draft Policy:	Waste Strategy Programme Team			
	Deputy Leader and Cabinet Member for			
	Environment and Carbon Reduction			
Did the Policy go to a Scrutiny	Yes ⊠ No □			
Committee Meeting:				
	05/02/2025			
Governance				
The Policy applies to:	All borough householders and those who use the			
	Council's waste and recycling collection services			
The Policy is available in the	Publications Scheme			
following locations:	Intranet / SharePoint ⊠			
	Council website / Service webpage			
This Policy has related documents:	Yes ⊠ No □			
•				



	Waste Strategy (2020-25), Environment Act (2021), Government Resources and Waste Strategy (2018) Climate Change Policy (2021-24)
The Policy was disseminated to:	
This Policy is due its next review:	01/04/2026

Document Control					
Date	Version	Action	Amendment(s)		
July 2023	Draft 1.1	Moved to new			
		template			
February 2023	Draft 1.2	Feedback	Multiple changes		
	incorporated ready		following further		
		for scrutiny	decisions on service		
			models		
December 2024	Draft 1.3	Updates to align to	Multiple changes		
		operational service	following further		
		change	decisions on service		
			models		

# 1 Contents

1	(	Cor	ntents	1
2	F	or	eword	3
3	I	ntr	oduction	3
4	"	Po	licy on a Page' (POP)	1
5	F	Pol	icy Statement	2
6	F	Pol	icy Context	2
	6.1	1	National Context	2
	6.2	2	Local Context	3
	6.3	3	Definitions	4
	6.4	4	What does the Policy seek to achieve?	4
7	٧		ste and Recycling Collection Policy Body	
Th	is	ро	licy covers the following key areas:	5
	7.1	1	Legislative background	6
	7.2 nro		Method of collections and container provision – for standard kerbside collection	7



	7.3	Presentation of household waste – for standard kerbside collection properties	11
	7.4	Contamination and unauthorised waste	11
	7.5	Additional Waste and Recycling Capacity	12
	7.6	Excess waste	13
	7.7	Missed collections	14
	7.8	Assisted collections.	15
	7.9	Service disruption	16
	7.10	Garden Waste Collection Subscription Service	17
	7.11	Bulky household waste collections	18
	7.12	Clinical waste	19
	7.13	Offensive Waste and Incontinence Waste	21
	7.14	Properties with restricted access	21
	7.15	Flats or properties with communal shared bins	22
	7.16	Multi-occupied properties	25
	7.17	Places of religious worship	25
	7.18	Premises occupied by a charity	26
	7.19	Waste collection from community centres and village halls	26
	7.20	Properties of mixed use (business and residential)	26
	7.21	Residential properties also used for business	27
	7.22	Collections from schools and education establishments	28
	7.23	Non-domestic and commercial / business waste	28
	7.24	Collections from outdoor events	28
	7.25	Communication, Education and Enforcement protocol for waste and recycling	
		ces	
	7.26	New developments and planning guidance	
8		at evidence have you used that suggests your Policy approach is the right one?	
9	Wh	o or what will be affected by the Policy?	
	9.1	Inclusion and Diversity	
1	O Cor	porate Knowledge	31
1	1 Acc	companying Strategy	32
1	2 I inl	s to other Corporate Policies or Partner documents	32



13	Apper	ndices			 	32
1	3.1	Glossary of te	erms and condit	ions	 	32

#### 2 Foreword

Our Waste and Recycling service is one of the key services we deliver for our residents it is the one point of contact for everybody who lives within the Borough. It plays an essential role in keeping our neighbourhoods and open spaces clean and tidy. It also helps to instil a sense of pride in where we live and our Waste and collection Recycling operatives incredibly hard in all weathers to make sure that this service is delivered to the best standard possible.

Delivering a Waste and Recycling service which meets the needs of our Borough and it's residents and helps to encourage positive environmental behaviours is of the utmost importance to Basildon Council.

This policy outlines the way in which Basildon Council will support its residents and businesses to contribute to achieving these aims and puts the needs of our Borough at their centre.



Councillor McGurran, Cabinet Member for Environment and Leisure

#### 3 Introduction

All Borough householders receive a waste and recycling collection service. This policy covers Basildon Borough Council services related to waste and recycling collections and sets out the responsibilities of the householder and those who use the Council's waste and recycling collection services when presenting their waste and recycling for collection, and the Council's responsibilities for collecting it.



This policy sits alongside the Council's Waste Strategy and supports our <u>Waste Strategy</u> ambition:

"For Basildon to have an efficient and environmentally sound waste service that delivers for householders and businesses".

### 4 'Policy on a Page' (POP)



# Waste and Recycling Collection Policy

## **Policy Statement**

Basildon Council commits to ensuring our waste and recycling collection service adheres to the requirements of legislation, including national and local targets, whilst also seeking to improve customer satisfaction and promote transparency of the responsibilities of both the Council and householders in respect of waste management.

# What does the policy seek to achieve?

Outcome: To deliver the Council's Waste Strategy, aiming to improve waste and recycling performance and meet governmental legislation.

It aims to:

- To set out waste and recycling responsibility and expectations for both householders, businesses and Basildon Borough Council
- Ensure there are robust practices and mechanisms are in place to deliver an efficient and effective waste and recycling service
- To enable and introduce measures to minimise waste and drive-up recycling rates

If you have questions about how to use the waste and recycling services, the policy will provide the answers, including:

- What waste and recycling services we provide
- How we collect waste and recycling from different types of property
- How you should present your waste and recycling materials
- What to do if something goes wrong, such as a missed collection



How and when we educate and enforce

Other related policies and strategies:

Waste Strategy (2020-25),
Climate Change Policy (2021-24)
Government Resources and Waste Strategy (2018)
Environment Act (2021),

#### 5 Policy Statement

Basildon Council commits to ensuring our waste and recycling collection service adheres to the requirements of legislation, including national and local targets, whilst also seeking to improve customer satisfaction and promote transparency of the responsibilities of both the Council, householders and other users of Basildon Council's services in respect of waste management.

### **6 Policy Context**

#### 6.1 National Context

How waste is collected and managed is in a period of change. This change was initiated by the Government's Waste Strategy (2018) which set specific goals to:

- Minimise waste and promote resource efficiency.
- Move toward a circular economy, encouraging reuse and repair.
- Significant changes to waste and recycling collections nationally with the introduction of the Government's Environment Act (2021) and to ensure services are provided which meet the requirements of the 'Simpler Recycling' policy

The Environment Act (2021) was born out of the strategy, with the aim to meet the goals above. It sets out legislation of how waste and recycling must be collected within specific parameters, with the key points included in the body of the policy (section 4.1)

Simpler recycling is government policy that simplifies the rules to make recycling easier for people and businesses within England. This allows for more streamlined collections from all households, businesses and relevant non-domestic premises (such as schools and hospitals).

Local authorities and other waste collectors will be able to co-collect some waste streams by default meaning that they will no longer need to collect 7 separate streams. The new default requirement for most households and workplaces will be 4 containers for:



- residual (non-recyclable) waste
- food waste (mixed with garden waste if appropriate)
- paper and card
- all other dry recyclable materials (plastic, metal and glass)

These may be various container types, including bags, bins or stackable boxes.

Weekly food waste collections will be mandatory by 2026, with micro-businesses given until 2027 to comply. The policy also encourages flexibility for local authorities to tailor services while maintaining national consistency. Full implementation is planned by 2027.

#### 6.2 Local Context

Due to the anticipated national changes, Basildon Borough Council introduced a Waste Strategy (2020-25). The strategy aims to bring the service into the 21st Century and meet the following strategic goals:

- 1. Driven by data we understand the borough and use our knowledge to make decisions based on data.
- 2. Overhaul our operations we deliver efficient and modern services we can be proud of.
- 3. Enhance the environment We do the right thing for your local area and the wider environment.

Historically, Basildon Borough has generated the highest amount of waste per household per year in Essex, equating to nearly one tonne per property.

Essex County Council have developed an Essex Waste Strategy which Basildon Council has adopted. This sets out the longer terms vision for managing all of Essex's Waste over the next 30 years.

Basildon Council has adopted the Essex Waste Strategy, developed by Essex County Council, which outlines the long-term vision for managing waste across Essex from 2024 to 2054.

This strategy includes several key targets and ambitions aimed at promoting sustainable waste management and meeting climate goals. One priority is achieving net zero greenhouse gas emissions by 2050. Waste reduction targets include halving residual waste per person by 2042. Recycling efforts to reach for at least 65% of waste to be reused, recycled, or composted by 2035. Waste disposal targets include sending no



more than 10% of waste to landfill by 2035. All Essex residents will have access to separate food waste collections by 2026 and access to comprehensive recycling service for plastics, paper card, metal, glass food and garden waste by 2026.

On top of this, it has a declining recycling rate and in 2021/22 it was 44%, where it has fallen from as high as 53% in 2014/15. This has to change to meet legislation and the commitment of achieving a 55% recycling rate by 2025 and 65% by 2035.

The Council are making changes to collection services over the next few years and the introduction of waste and recycling collection policy allows for these changes to be implemented successfully.

#### 6.3 Definitions

Glossary of terms and definitions can be found at the end of the policy document in section 13.1.

#### 6.4 What does the Policy seek to achieve?

Outcome: To deliver the Council's Waste Strategy, aiming to improve waste and

recycling services for the community and encouraging residents to take

pride in their local area.

- Efficient Waste Services: Provide waste services that meet the community's needs, ensuring that waste collection, recycling, and disposal are carried out efficiently and effectively.
- 2. **Waste Reduction**: Encourage the community to reduce waste generation through education and initiatives that promote sustainable practices.
- 3. **Resource Recovery**: Improve resource recovery by increasing the rates of recycling and reusing materials, thereby diverting waste from landfills.
- 4. **Environmental Protection**: Minimise the environmental impact of waste management activities by reducing pollution and promoting the use of environmentally friendly technologies.
- 5. **Compliance with Legislation**: Ensure that all waste management practices comply with local, national, and international regulations and standards.



- 6. **Economic Efficiency**: Implement cost-effective waste management strategies that provide economic benefits, such as reducing disposal costs and creating opportunities for resource recovery.
- 7. **Long-term Planning**: Develop and maintain long-term waste management plans that anticipate future needs and challenges, ensuring sustainable waste management for the future.
- 8. **Community Engagement**: Foster community involvement and support for waste management initiatives through education, awareness campaigns, and public consultation

# 7 Waste and Recycling Collection Policy Body

# This policy covers the following key areas:

- How the Council collects waste and recycling and what containers are provided – for standard kerbside collections (see section 7.2) or for properties that use communal bins (see section 7.15)
- How residents should present their waste (see section 7.3) and what happens if materials are presented incorrectly (see section 7.4 and 7.6)
- The provision of additional containers and how to request these (see section 7.5)
- What is classed as a missed collection and what to do if it happens (see section 7.7)
- How residents can request additional support via assisted collections (see section 7.8)
- How residents can find more details about certain collection services including the garden waste subscription service (7.10), the Bulky Waste bookable service (7.11), and the Clinical Waste collection service (7.12)
- How residents can find out how we collect waste from places of worship (7.17), charities (7.18), community centres or village halls (7.19), and schools (7.22)
- What residents should do with commercial / business waste (see section 7.23)
- How the service educates and enforces this policy (see section 7.25)



 What developers should consider during development of housing (see section 7.26)

#### 7.1 Legislative background

- 7.1.1 Basildon Borough Council is a waste collection authority with a statutory duty under the provisions of the Environmental Protection Act 1990 (as amended), in particular, Section 45 of the Act, to arrange for the collection of household waste in its area and the collection of commercial waste for which a charge can be made.
- 7.1.2 The statutory duty does not prescribe the method of collection of household waste. The Council, therefore, needs to determine the type and frequency of collections to be made available in the area and set out the policies to be applied in relation to these collections.
- 7.1.3 The Environment Act 2021 is an addition to the Environmental Protection Act 1990. The provisional amendments to section 45a of the Environmental Protection Act will impact the way in which Basildon Council collects waste and recycling by requiring:
  - a) Recyclable household waste to be collected separately from other household waste and must be collected for recycling or composting
  - b) Different recyclable household materials to be collected separately where appropriate, 2 or more recyclable materials can be collected together, but only if economically and environmentally advantageous.
  - c) Food waste must be collected weekly and separately
- 7.1.4 Simpler Recycling is a government policy in England designed to streamline waste management for households, businesses, and non-domestic premises. It enables more efficient collection by allowing co-collection of certain waste streams, reducing the need for seven separate collections. Weekly food waste collections will be mandatory by 2026, with micro-businesses given until 2027 to comply. The policy balances national consistency with local flexibility, allowing authorities to tailor services to community needs while promoting efficient and sustainable recycling.
- 7.1.5 The Collection Policy is published in accordance with the provisions of the Environmental Protection Act 1990 (EPA) which enables the Council to:
  - a) Specify the type of container to be used by the householder for the collection of their non-recyclable waste
  - b) Specify the type of containers to be used by the householder for the waste which is to be recycled or composted



- c) Specify the size, construction and maintenance of the containers provided
- d) Determine the position that householders should place their containers for emptying by the Council and the steps to be taken by householders to facilitate the collection of waste from the containers
- e) Take enforcement action against a householder who fails, without reasonable excuse, to comply with the Council's requirements under this legislation
- f) Make a charge to service users for the provision of waste containers, if it so wishes
- 7.1.6 The Controlled Waste (England and Wales) Regulations 2012, in particular Schedule 1, defines what waste is to be treated as household waste or otherwise the waste that is considered to be industrial or commercial waste. These Regulations also define the types of household waste for which collection and / or disposal charges may be made.

# 7.2 Method of collections and container provision – for standard kerbside collection properties

- 7.2.1 Basildon Borough Council operate a kerbside separated collection system for residents of the borough. This requires materials to be separated into the different waste streams ready for collection. This helps ensure that as much waste material as possible can be recycled or composted and increases the quality of the recyclable material presented by reducing contamination.
- 7.2.2 The Council's website and the Our Streets App is updated periodically, which sets out which materials are to be placed in each container.
- 7.2.3 Separate collection arrangements may exist for those people living in flats who make use of communal collection services. You can find further details in section 7.15 regarding waste collections from flats or communal bins.
- 7.2.4 In accordance with the Environmental Protection Act 1990, Basildon Borough Council directs householders receiving individual kerbside collections to use the container(s) provided and to present the materials for collection as described below (latest information can be found at <a href="https://www.basildon.gov.uk/waste-information">www.basildon.gov.uk/waste-information</a>):

Material	Container	Collection Frequency
Food Waste	<ul> <li>a) Grey 5 litre internal kitchen caddy and</li> <li>b) Green 23 litre outdoor caddy</li> <li>c) One roll of 50 compost caddy</li> </ul>	Weekly



	liners supplied to each household per year.	
Garden Waste Subscription Service (Opt-in)	Green 240 Litre Wheeled Bins with permit sticker attached.	Weekly – must be subscribed and have permit sticker
Non-Recyclable Waste	180 Litre Wheeled Bins (black sacks for properties not operationally suitable for wheeled bins)	Weekly
Recycling materials (except glass, textiles and electrical)	2 x single use bags (1 for paper and card, 1 for plastics and cans)	Weekly
Glass bottles and jars	55 litre orange box	Once per fortnight
Textiles	Textiles Sack	Fortnightly – collected alongside glass collections by third party contractor
Small electrical items	Householders are able to drop these off at local recycling centres and libraries	During open hours

- 7.2.5 All containers supplied by the Council remain the property of the Council and should not be removed from the household address to which they have been issued. Householders are encouraged to appropriately mark their containers and collection containers with their house number or property name so that they can readily identify them. Any markings deemed offensive will result in the containers being replaced. The Council may choose to charge the resident for such replacement in line with the EPA 1990 legislation. Permanent markings should not be used in order that the Council can reissue or recycle containers where applicable, in line with the Council's strategic goal of enhancing the environment through recycling.
- 7.2.6 Householders are required to keep and maintain the containers provided to them by the Council in a safe and clean condition. The Council do not provide a container cleaning service. Householders should only use the container for the intended purpose, i.e., the presentation of materials for recycling, composting and waste for collection. If the above is not met, the Council may undertake education and / or enforcement action (see section 7.25).



- 7.2.7 Householders should only use the containers issued to that property. Unauthorised use of containers by householders may result in education and/or enforcement action being taken. However, for paid services such as garden waste collections, shared use is allowed provided it is agreed by the subscribing resident.
- 7.2.8 The Council will only collect waste and recycling materials from containers provided by the Council with the exception of properties who are required to use black sacks for non-recyclable waste collections. Loose cardboard that cannot fit in the container can be placed out alongside the container providing it is flattened down and does not contain and other materials.
- 7.2.9 The Council cannot be held responsible for any damage to property or persons caused by containers left at roadside, for example if the damage is caused by extreme weather or if moved by a third party.
- 7.2.10 The Council will supply kerbside containers and deliver these to the edge of your property boundary when requested via the <a href="www.basildon.gov.uk/whereilive">www.basildon.gov.uk/whereilive</a> based on the following:

Container	Free if:	Charged if:
Grey 5 litre internal	New dwellings (where suitable	Damaged by householders
kitchen caddy and	containers are not present),	due to negligence, or if
Green 23 litre outdoor caddy	damaged, stolen, additional	they are deemed excessive requests
Kitchen Caddy Food Liners	Roll of 50 delivered annually	Additional liners can be purchased at local libraries or Basildon Centre
55 litre orange box	New dwellings (where suitable	Damaged by householders
	containers are not present),	due to negligence or if they
	damaged, stolen, additional	are deemed excessive
		requests
180 Litre wheeled	New dwellings (where suitable	Damaged by householders
Bins – non-recyclable	containers are not present),	due to negligence or if they
waste	damaged, stolen, meet eligibility	are deemed excessive
	criteria for additional containers (see	requests
	section 7.5)	
Single use bags –	New dwellings	Damaged by householders
1 x for paper and card	2 x rolls of each bag delivered	due to negligence or if they
1 x for plastics and	annually. Each roll contains 26 sacks	are deemed excessive
cans	A LPC and allower than a control of	requests
(approx. 90litres each)	Additional rolls of bags can be	
	obtained by householders directly	
	from Barleylands Depot, the Basildon	
	Centre or local Libraries.	



Green 240 litre	Garden bins will be provided	Damaged by householders
wheeled bin – garden	following a subscription (if containers	due to negligence, or if
waste (if subscribed to	are not already present), or if they	they are deemed excessive
the service)	are damaged or stolen while	requests
,	subscribed	

- 7.2.11 The Council can charge at full-cost recovery for containers, including unit and delivery costs and may apply a charge to containers (in accordance with the EPA 1990) that are typically categorised as free, if the requests are excessive, or on an unreasonable basis i.e., if being stolen / going missing regularly or containers are damaged by the resident through misuse.
- 7.2.12 The Council will require householders to exchange their damaged container when receiving their new container, by leaving the container at your usual collection point on the day of delivery of the new container.
- 7.2.13 The Council will provide, free of charge, a full set of containers (with the exception of a green wheeled bin for garden waste), as appropriate, to new dwellings within two weeks of occupation, provided that at least four weeks prior notice of the occupation of any new property is provided by the developer, property owner, landlord, managing agent or the new householder.
- 7.2.14 The Council will provide free of charge a roll of 50 food liners to each household on an annual basis, residents will be informed ahead of planned delivery via the Council's website.
- 7.2.15 Some properties may not be suitable for specific types of containers, these would typically be properties that are flats, properties that are hard to reach or far from the highway If you require container or service adjustments, please contact <a href="mailto:recycling@basildon.gov.uk">recycling@basildon.gov.uk</a> and we will consider your request on a case-by-case basis. For example, here are some reasonable adjustments:
  - > Retain black sack collections.
  - Receive different size containers.
  - Communal bins may be provided instead of kerbside containers (see section 7.15)
  - ➤ The above is a guide and each property that requests this is decided on a caseby-case basis. The Council has final say on deciding on the appropriate service being provided.
- 7.2.16 Some properties may require adjusted collection frequency. These would typically be properties that are flats or those with communal bins. The Council will deem frequency of collections for these types of properties. For further information (see section 7.15)



#### 7.3 Presentation of household waste – for standard kerbside collection properties

- 7.3.1 Materials to be collected must be presented for collection in the appropriate container at the edge of your property, or at an agreed collection point by the householder by 7:30am on the day of collection. Containers must not block access, pathways, roads or cause trip hazards for those who are visually impaired. Waste presented later than 7:30am does not count as a missed collection and will not be collected under the missed collection policy (see section 7.7).
- 7.3.2 Containers once emptied will be placed at the collection point for the householder to then remove them from the kerbside [or any other part of the footpath or highway] as soon as practicable on the same day of collection. Containers must not be left on the footpath or highway between collections.
- 7.3.3 If householders do not present their containers correctly, and / or fail to remove them from the kerbside on the day of collection then they may be subject to education and / or enforcement action (see section 7.25)
- 7.3.4 If the placing of containers at the edge of your property, in the front or rear of their property causes an obstruction or difficulty in collection the Council may agree an alternative collection point with the householder.
- 7.3.5 Where the Council undertakes collections by passing over a private road or drive, the property owner will be required to sign a waiver to say that they will not claim against the Council for any liability for damage to the road or other surfaces because of the collections. If such a waiver is not signed, householders will be required to present their containers at an agreed collection point that can be accessed directly from the adopted public highway.
- 7.3.6 Collection crews will clear up spilled waste near the vicinity where is safe to do so of the collection point. If significant it can be reported via the Council website at <a href="https://www.basildon.gov.uk/street">https://www.basildon.gov.uk/street</a>. Any spillage caused by householders' inappropriate actions when presenting their household waste for collection may be subject to education and / or enforcement action (see section 7.25)
- 7.3.7 The Council will only empty containers that contain the items stipulated for each type of container as listed on the Council's website at <a href="https://www.basildon.gov.uk/Waste-information">https://www.basildon.gov.uk/Waste-information</a>.
- 7.3.8 Containers with the wrong material inside will be dealt with under the contaminated container policy (see section 7.4)

#### 7.4 Contamination and unauthorised waste

7.4.1 Contamination is defined as waste materials being placed for collection in the wrong container(s)



7.4.2 Contamination can be caused when non-recyclable waste is placed in a container for recycling materials or when recyclable waste is placed in a container for non-recyclable waste material.

The Council will only empty containers that contain the items stipulated for each type of container as listed on the Council's website at <a href="https://www.basildon.gov.uk/Waste-information">https://www.basildon.gov.uk/Waste-information</a>

- 7.4.3 If a container is contaminated it will not be emptied, a contaminated sticker/bin hanger will be left on the container. In such instances the householder will need to remove the contaminated material and re-present the container, uncontaminated, on the next scheduled collection day. Contamination may lead to education and enforcement action.
- 7.4.4 The Council will not return to empty a contaminated container under the missed collection policy (sec 7.7).
- 7.4.5 Failure to remove contaminated materials from the container may result in the Council taking education and / or enforcement action against the householder (see section 7.25). If persistent misuse of collections occurs, the container[s] may be removed.
- 7.4.6 If the householder does not remove the contaminated material, they will need to make their own arrangements for disposal of this waste at their own expense.
- 7.4.7 Unauthorised waste material, such as hazardous or clinical waste (see section 7.12) in any container will also be classed as contamination. A consequence of this will be that the entire contents of the container will not be collected. Householders will be required to safely remove the wrong items from the container and present it correctly on the next collection day, this can not be reported as a missed collection. Information on what goes in each bin can be found on Basildon Council's Waste and Recycling webpage, <a href="https://www.basildon.gov.uk/Waste-information">https://www.basildon.gov.uk/Waste-information</a>.
- 7.4.8 Hazardous waste such as asbestos, chemicals or gas canisters will not be collected. To find out how to dispose of hazardous materials via Essex County Council go to <a href="https://www.loveessex.org/news-and-ideas/household-hazardous-waste-in-essex/">https://www.loveessex.org/news-and-ideas/household-hazardous-waste-in-essex/</a>
- 7.4.9 Householders using domestic waste containers for business or commercial waste will not be collected and may be liable to prosecution, if that use contravenes the Controlled Waste (England and Wales) Regulations 2012, or any other relevant legislation. This will be subject to education and / or enforcement action (see section 7.25)
- 7.5 Additional Waste and Recycling Capacity



- 7.5.1 The Council at its own discretion, are able to provide additional recycling containers upon request where the request is not seen as excessive in regard to the size of the property. See section 7.2.10 for details on criteria to receive additional containers and whether there would be a charge for them. 7.2.4
- 7.5.2 The Council will be able to provide additional non-recyclable waste capacity on request if they meet the following eligibility criteria. A waste audit may be completed by the waste and recycling service to ensure that you meet the criteria:
- a) You already recycle everything you can, and
- b) The household produces high levels of non-recyclable waste this could include offensive waste such as nappies or incontinence waste
- 7.5.3 Applications for additional non-recyclable waste capacity will be reviewed on an individual basis.
- 7.5.4 Those in receipt of additional non-recyclable waste capacity may be subject to review by the service. It is the responsibility of the householder to inform the Council of any changes to their circumstances that relate to an approved application to receive additional non-recyclable containers.

#### 7.6 Excess waste

- 7.6.1 The Council will collect all household recycling provided that any such materials are properly presented and clearly identifiable on the scheduled collection day, material that is secure and free from contamination. To receive collection of garden waste householders must subscribe to this service www.basildon.gov.uk/gardenwaste
- 7.6.2 In the case of collections from wheeled bins, both non-recyclable waste and garden waste, wheeled bin lids should be closed for collection, not overfilled, or too heavy.
- 7.6.3 The Council will not collect any side waste for non-recyclable or garden waste collections, i.e., any waste outside of the allocated wheeled bin.
- 7.6.4 The Council will collect side waste for dry-recyclable collections, when presented in a safe and tidy way, and free from contamination. For example, cardboard must be flattened and stacked prior to collection.
- 7.6.5 Should excessive levels of certain materials be presented for collection this could result in a container not being collected, this is based on the discretion of the collection crew. For example, a small amount of animal waste would be acceptable, but several bags of such material would be deemed excessive, hazardous and not typical of a regular household waste and requiring of specialist treatment.



- 7.6.6 Waste that is not properly contained or presented as directed may constitute a littering or fly-tipping offence (with the exception of split sacks as defined in section 7.3.6) and may result in education and / or enforcement action being taken against the householder deemed responsible. Waste remains the responsibility of the householder until it is collected by the Council.
- 7.6.7 Excessively heavy containers will not be emptied, where the crew cannot safely empty the container. In these circumstances the householder will be required to remove sufficient material from the container in order that it can be safely emptied on the next scheduled collection day.

#### 7.7 Missed collections.

- 7.7.1 The Council will use reasonable endeavours to empty all containers presented for collection on the published scheduled day of collection for that property address. Missed collections must be reported to the Council from 4pm on the day it was missed and for 24hrs after. Following a report of a missed collection the Council will endeavour to complete the collection within three collection working days after the report. However, if the Council is unable to do this, householders will be required to take their container back into their property until their next scheduled collection day.
- 7.7.2 The Council will only return to collect a missed collection which has been deemed a valid request. The Council will not return to collect where:
  - It has been determined that the container(s) were not presented correctly see section 7.3. Typically, this includes, container(s) not being presented by 7:30am (which may be evidenced using the Council's in-cab digital system) at the collection point, wheeled bin lids not being closed, or containers not in the agreed location / collection point.
  - It has been determined that the containers have been contaminated or contain unauthorised waste – see section 7.4
  - Waste is deemed side waste, i.e., waste for non-recyclable or garden waste outside of the provided containers— see section 7.6 for further guidance
  - Waste is deemed excessive i.e., containers are overly compacted and cannot be fully emptied, or are too heavy to safely manoeuvre see section 7.6
  - The container(s) have already been confirmed as being collected and emptied once on the scheduled day of collection.
  - Safe access to remove waste/recycling without damaging property is not available
  - Abuse/Mistreatment towards collection crews prevented safe collection taking place
  - Adverse weather has prevented the safe collection of household waste



- 7.7.3 If the container(s) is not collected due to the above reasons this will be recorded as a justified non-collection and will not be recorded as a missed collection. In these circumstances the containers will be emptied on the next scheduled collection day, provided they are presented in accordance with the Waste and Recycling Collection Policy.
- 7.7.4 The Council will not return to empty containers where the missed collection is reported more than 24 hours after the scheduled collection day or where the reported missed collection is deemed to be not valid. The customer will be required to represent the items on the next appropriate scheduled collection date.
- 7.7.5 Should waste be found to be frozen within a container, reasonable endeavours will be made to return to the container should the temperature sufficiently raise to thaw the waste. If this does not happen before it is operationally feasible to return, this may be on the next scheduled collection date.

#### 7.8 Assisted collections.

- 7.8.1 The Council currently offers an assisted collection service to householders who find it difficult to move containers to the boundary of their property. A request for an assisted collection will only be considered where there is no person living at the household to help and:
  - > the householder has a disability,
  - the householder requires short term assistance due to an injury or operation,
  - > the householder is reliant on a carer who does not live at the property.
- 7.8.2 The Council defines an assisted collection as the collection of a bin or container by the collection crews from an agreed collection point within the curtilage / boundary of the property, returning the container(s) to the location after they have been emptied.
- 7.8.3 All assisted collection points will require approval by the Waste and Recycling service, this may require an inspection.
- 7.8.4 The agreed collection point should be outside and freely accessible without engagement with the householder; no pets will have access to the agreed collection point on the day of collection. The collection point should be as close to the adopted highway as is practically possible and in accordance with the needs of the resident. In agreeing the designated collection point consideration will be given by the Council to any health and safety risks associated with access onto the property, and an appropriate risk assessment may be completed.



- 7.8.5 Householders are required to apply to the Council for this service and sign a waiver that residents will not claim for any damage caused by operatives entering their property boundary to collect their waste. Under the Occupiers Liability Act, all householders have a duty of care to Council operatives.
- 7.8.6 Eligibility for this service arrangement will be reviewed periodically. If at any time the Council has reason to believe that the recipient no longer meets the criteria for the assisted collection service, or it does not meet health and safety standards, the provision of this service will be reviewed accordingly and may be removed.
- 7.8.7 Householders shall be required to notify the Council of any changes in their circumstances that may affect their eligibility for this collection or changes to property affecting access that pose a health and safety risk.
- 7.8.8 In the case of the Council refusing an application for an assisted collection service, the applicant will be provided with an explanation of the reason for the refusal. Collections will only be collected via the standard kerbside collection until an assisted collection service is approved by the Council.

#### 7.9 Service disruption

- 7.9.1 Where collections are disrupted due to Bank Holidays, particularly over the Christmas/ New Year period, the Council will put in place a revised collection schedule. Any such arrangement will be published in advance on the Council website, Our Streets App and social media pages.
- 7.9.2 During periods of severe weather (i.e., high winds, snow, or flooding), the Council will use reasonable endeavours to maintain scheduled waste collections. However, the Council may have no option other than to suspend collections on the grounds of safety or inability to access certain roads. In such circumstances, containers will be emptied on the next scheduled collection day following conditions being safe and subject to the required resources being available. The Council may choose to suspend certain collections to use the resource to help with catch-up of collections. If this is the case, it will be communicated on the Council website, Our Streets App and social media pages. If waste collections continue to be unsafe or unable to be delivered, it will be carried out as soon as possible.
- 7.9.3 Where general service disruption occurs only a partial collection can be achieved, priority of collections will be determined by the waste and recycling collection service:

Order of Suspended services:

- 1. Glass
- 2. Dry Recycling
- 3. Garden Waste
- 4. Residual Waste



#### 5. Food Waste

#### 7.10 Garden Waste Collection Subscription Service

- 7.10.1 To use the domestic Garden Waste Collection Service residents must pay each subscription year to subscribe to the service and have the required container and permit for garden waste to be collected from the property.
- 7.10.2 The service is only available to domestic premises, properties classified as a residential property within the Local Land Property Gazetteer (LLPG) and be located within the Basildon borough.
- 7.10.3 There are some properties in the Borough that the Council is unable to offer a garden waste service due to access issues for providing a wheeled bin collection. Those properties may be able to agree a suitable collection point if one is available, but this is down to the discretion of the Council.
- 7.10.4 For other properties such as schools, village halls and charities will have separate arrangements to subscribe to the service.
- 7.10.5 The subscription period runs from 1<sup>st</sup> April to 31<sup>st</sup> March annually. All subscriptions end on the 31<sup>st of</sup> March no matter the time of year the subscription begins.
- 7.10.6 The cost of the service is determined by the Council on an annual basis.

  Subscribers will be informed of any change to their annual subscription via the Council's website at <a href="https://www.basildon.gov.uk/gardenwaste">www.basildon.gov.uk/gardenwaste</a>
- 7.10.7 The service provided is as follows:
- ➤ A fixed subscription price per annum for the first bin for each household
- A maximum of ten bins per household
- ➤ The service is weekly, except for a set break during and following the Christmas period in which residents will be notified of this break in advance. Dates will be confirmed and communicated each year prior to the break taking place.
- 7.10.8 Residents must agree to the Terms & Conditions of the service when subscribing and resubscribing to the Council's Garden Waste Collection Service.
- 7.10.9 Following subscribing the resident will receive a bin if they do not already have one and a permit will be delivered in the post to affix to the bin.
- 7.10.10 The service will not empty the bin when presented for collection and may take the bin away if the permit sticker is no longer valid or if the permit is not displayed correctly:



- > The permit sticker must remain on the bin.
- ➤ The permit sticker attached to the bin must have the same address as to where the bin is being presented for collection.
- ➤ The permits Subscription Number is valid, associated to a paid for active subscription.
- ➤ The correct subscription year's permit sticker is attached to the bin the permit sticker is replaced every annual subscription.
- An original permit must be affixed to the bin, any incorrect or non-original permits presented on the bin at point of collection will not be emptied.
- 7.10.11 Only loose garden waste, as listed in the current acceptance list may be placed in the garden waste bin (<a href="https://www.basildon.gov.uk/gardenwaste">https://www.basildon.gov.uk/gardenwaste</a>). Waste should not be presented within bags of any form, even if compostable. Bins containing the wrong material fall under the contaminated bin policy (see sec 7.4)
- 7.10.12 Any garden waste collected must not include garden waste produced by gardeners, as this is classified as commercial waste. If the Council determine that the service is being used for commercial gain, by a gardener the service can be revoked with no refund
- 7.10.13 Garden wheeled bin lids should be closed when presented for collection, not overfilled or too heavy. The service reserves the right not to collect any wheeled bins if the wheeled bins lids are open or overfilled.
- 7.10.14 The full details of the service can be found at <a href="www.basildon.gov.uk/gardenwaste">www.basildon.gov.uk/gardenwaste</a> including the terms and conditions of the service.

#### 7.11 Bulky household waste collections

- 7.11.1 The Council can arrange for the collection of large items of household waste, such as furniture and electrical items, etc. from residential properties where these cannot be contained within the containers provided by the Council or where the item exceeds a certain weight where it can't be collected by the usual collection crew. A list of accepted items can be found on the council's website at <a href="https://www.basildon.gov.uk/bulkywaste">www.basildon.gov.uk/bulkywaste</a>.
- 7.11.2 These collections need to be arranged based on availability and items can only be collected from outside the property at the properties collection point. A collection day will be specified, but not an exact time during the day. Items must be out for collection by 7.30 on the specified day of collection. Only items listed in the booking will be collected. Additional items presented will not be collected. The Council reserves the right not to collect any items deemed inappropriate or considered to be unauthorised waste (such as commercial waste) or items that pose a health and safety risk to collection crews.



- 7.11.3 A collection and administration charge are made for this service and this charge must be paid in advance of the bulky item[s] being collected.
- 7.11.4 Refunds can't be issued once a booking is made as long as it is 2 days prior to the scheduled day of collection.
- 7.11.5 Any amendments to the booking, or cancellation must be made a minimum of 48hours prior to the collection slot.
- 7.11.6 No refund will be paid based on the following:
  - ➤ The required notice (2 working days prior to the scheduled collection) is provided.
  - ➤ If the Council miss the bulky waste collection we will return within 2 working days following the resident reporting it has been missed (see section 7.7)
  - If the items presented are contaminated or could cause a health hazard to the collection crew.
  - ➤ If the Council suspends the service in exceptional circumstances (see section 7.9) as the we will return when normal service is resumed
  - If any materials are collected by a third party not working on behalf of the council, with or without the resident's approval.
- 7.11.7 Bulky wastes deposited on the Highway without making adequate provision for their safe removal and disposal will be treated as fly-tipping and will be investigated by the Council's Enforcement team and may result in a fine/legal action.
- 7.11.8 If the Council misses a bulky waste collection which was deemed to be presented correctly, the Council, on notification, will use reasonable endeavours to return to make the bulky waste collection within two working days. The Council will not issue refunds for missed collections (see section 7.7 for more details on missed collections)
- 7.11.9 The Council aims to divert as much waste to be reused, repaired, and recycled, by working with partners and promoting companies that undertake this work, as well as sorting materials before disposing. Visit the Council's website for more information.

#### 7.12 Clinical waste

- 7.12.1 Clinical waste is medical waste produced from healthcare or similar activities that may pose a risk of infection. Clinical waste can be hazardous to anyone who comes into contact with it. It can include:
  - > Human or animal tissue



- ➤ Blood or other body fluids
- ➤ Excretions
- > Drugs or other pharmaceutical products
- ➤ Used swabs or dressings
- ➤ Used needles, syringes or other sharp instruments
- 7.12.2 The Council has a duty to collect healthcare and clinical waste from domestic properties, conditional upon the following circumstances:
  - ➤ If patients treat themselves in their own home any waste produced because of treatment is their own.
  - ➤ In the case of pharmaceuticals (medicines etc.), the recommended means of disposal is to return them to a pharmacist. If this is not possible the Council is obliged to collect the waste separately when requested to do so by the resident
- 7.12.3 If patients are treated in their home by a community or district nurse or a member of the NHS profession, any resulting waste produced is considered to be the healthcare professional's waste. If the waste is classified as hazardous, the healthcare professional should remove that waste and transport it in approved containers to the healthcare provider's base for appropriate disposal.
- 7.12.4 Under the controlled waste regulations, the Council may charge for the collection of specific waste streams, including healthcare and clinical waste. However, the Council currently arranges for the collection and disposal of needles, other sharps and hazardous healthcare waste clinical from residential properties within the Basildon area free of charge. The Council, however, reserves the right to introduce a charge for collection of clinical and healthcare waste in accordance with the Controlled Waste (England and Wales) Regulations 2012. You can find out more and sign up to the service here <a href="https://www.basildon.gov.uk/clinicalwaste">https://www.basildon.gov.uk/clinicalwaste</a>
- 7.12.5 Syringes, needles or other sharps will only be collected if they are placed in a prescription sharps container. Empty sharps containers can be obtained from a GP surgery or other healthcare providers via a prescription.
- 7.12.6 Service users can also be provided with clinical waste sacks by the Council or Contractor for non-sharp related objects.
- 7.12.7 Householders **must not** dispose of syringes, needles and other hazardous healthcare wastes in any other waste or recycling container other than a prescription sharps container.
- 7.12.8 Service users shall be required to notify the Council of any changes in their circumstances that may affect their eligibility for this collection.



#### 7.13 Offensive Waste and Incontinence Waste

- 7.13.1 Offensive waste is defined as non-infectious waste, which is unpleasant and may cause offence to those coming into contact. It includes, human hygiene or sanitary waste, including nappies and incontinence pads. It also includes outer dressings and protective clothing, e.g., masks, gowns and gloves that are not contaminated with body fluids. These items can usually be disposed of as general non-recyclable waste.
- 7.13.2 This type of waste must be double wrapped prior to disposal to reduce any safety risk to the collection crews. The Council reserves the right not to collect any items if not appropriately presented for collection.

#### 7.14 Properties with restricted access

- 7.14.1 In situations where safe, efficient and economic collections cannot be made, for example the presence of steps or slopes that make manoeuvring containers hazardous, it may be necessary for the Council to specify alternative storage and collection arrangements for the property. This could be a communal bin location or a separate collection service. In determining the collection points for those affected properties, consultation will take place with the householders and / or managing agents concerned.
- 7.14.2 Where access to a property is controlled by electronic gates or other security barriers householders or their managing agent need to accommodate the arrival of the collection crews and provide timely entry, this could be by providing a key code or access fob. If access is not permitted within five minutes of arrival, the collection[s] will not be made and will take place at the next scheduled collection day. In this circumstance this is not classified as a missed collection.
- 7.14.3 Where access is not permitted to gated properties or properties that can only be accessed from an un-adopted road, householders will be required to present their containers outside the gates or at the edge of the public highway for collection at a location agreed by the Council. Should the premises be inaccessible, the property owner will need to make alternative arrangements for the collection of household wastes.
- 7.14.4 There are households within the Council area, particularly outlying rural properties, where the use of regular collection vehicles is impractical. In such circumstances the Council will use a smaller collection vehicle, and this may impact on the number of recycling services that are offered to that property.
- 7.14.5 To ensure that the Council retains an efficient and expedient level of service, all containers should be presented by the householder at an agreed collection point



- which will normally be where the end of the private road, driveway or other part of the property meets the public highway.
- 7.14.6 Where development of new properties is still taking place and roads are not yet adopted, but householders are in occupation, the Council will carry out a risk assessment to determine whether it is safe to enter the site to make collections. Where it is deemed unacceptable to make collections due to a health and safety risk, the Council will work with the developer to agree a temporary communal collection point. The waste collection crews will only make collections from this location once a satisfactory risk assessment is in place. The developer will be responsible for informing householders about the temporary arrangements. It will be the householders' responsibility to ensure that their waste/recycling is in the temporary area ready for collection by no later than 07.30 hours on the scheduled collection day.
- 7.14.7 Following completion of a new development an inspection may be carried out to determine appropriate collection points for each property and the developer will be responsible for informing residents of the new permanent collection points for their waste and recycling collection services.

#### 7.15 Flats or properties with communal shared bins

- 7.15.1 Some properties in the borough dispose of their waste using communal shared bins, which are bins used by multiple properties.
- 7.15.2 Blocks of flats will usually be provided with communal recycling and waste collection facilities, although in some circumstances, such as independent flats above commercial premises, maisonettes and similar, an individual kerbside collection may be provided.
- 7.15.3 Whilst the Council has an obligation to collect household waste, property owners, landlords and managing agents have a 'Duty of Care' obligation, imposed under section 34 of the Environmental Protection Act 1990 to ensure that all waste arising from the premises is:
  - ➤ safely and securely stored
  - > prevented from escaping from the property owners, landlords or managing agent's control
  - > prevented from causing environmental pollution or harming anyone
  - > Only passed to an appropriately licenced person for transfer and disposal
- 7.15.4 Failure to comply with their duty of care obligations is an offence and could lead to prosecution by the Council.



- 7.15.5 Communal collection services provided to blocks of flats and other properties will often need to be individually assessed and tailored to a specific location, taking into consideration:
  - > The number and type of property
  - > Bin and container storage capacity
  - > The presence of waste chutes
  - Any limitations on accessing a collection point for the collection crew and their vehicles.
- 7.15.6 The standard collection services for blocks of flats will comprise of the following:
  - > a weekly separate collection of food waste
  - > a weekly separate collection of recycling materials
  - > a fortnightly collection of glass materials
  - > a weekly collection of non-recyclable waste

However, these will be individually assessed and subject to change.

- 7.15.7 The number and capacity of the communal bins provided will be based on a formula calculation based on a standard collection capacity per dwellings / bedrooms and will vary according to the number and types of property serviced. However, in some circumstances this will need to be adjusted due to the capacity of any bin stores / compound areas.
- 7.15.8 It is the responsibility of the property owner, landlord or managing agent to manage new and existing bin storage areas. New bin storage areas must be signed off as part of the planning process for new/extended properties. Existing storage areas will be regularly reviewed and need to meet the following criteria:
  - Secure and convenient to encourage their responsible use by householders.
  - Large enough to allow each bin to be removed from the area without the need to remove other bins.
  - Be a hard standing surface area, and doors that can be secured as well as be opened to allow easy removal of any bins.

If bin storage areas do not meet the below criteria, the Council will not be liable to any damage caused by operatives.

- 7.15.9 If a property requires more frequent collections, the Council reserves the right to charge for the collection.
- 7.15.10 Where householders use a chute system for the disposal of their general non-recyclable waste, the property owner, landlord or managing agent will be required to manage the bin store area to prevent waste overspill.



- 7.15.11 Occupiers of flats of whatever tenure are required to present their waste in the manner prescribed by the Council using the bins and containers provided by the Council and in accordance with this policy.
- 7.15.12 The Council recommends that conditions should be included in any tenancy / leaseholder agreements to ensure that householders commit to segregating their waste for recycling and presenting it in the prescribed manner.
- 7.15.13 If the designated collection point is within the grounds of a property, it is the responsibility of the property owner, landlord or managing agent to arrange appropriate access before collections can be made. If access is blocked for any reason, including by designated parking spaces then the collection will not be made and will not be collected until the next scheduled collection. In this circumstance this is not classified as a missed collection.
- 7.15.14 The property owner, landlord or managing agent shall be required to purchase, keep and maintain the bins provided by the Council in a safe and clean condition and ensure that householders of the flats only use the bins and containers for their intended purpose. The Council are available for the property owner, landlord or managing agent to be contracted to clean the communal bins or if new communal bins are required. They will also need to contact the Council directly for new or additional container requests there will be a charge, unless damaged by operatives.
- 7.15.15 The Council will only collect waste from communal bins that is properly contained in the bin(s) provided and not on the floor. The Council will not empty bins that are overfilled or contaminated by unauthorised waste.
- 7.15.16 Where excess waste is left in bin stores and compounds, and this prevents access to bins and containers these will not be emptied. Any wastes, spillages that may attract vermin should be cleared by the landlord, property manager, or managing agency as a matter of urgency or may lead to bins not being emptied due to the health and safety risk. It may lead to enforcement action.
- 7.15.17 The Council will not remove bulky household waste items deposited in communal bin storage areas. This remains the responsibility of the property owner, landlord or managing agent. Residents living in flats are able to book bulky household waste collections at <a href="www.basildon.gov.uk/bulkywaste">www.basildon.gov.uk/bulkywaste</a> for which there will be a charge for collection and disposal.
- 7.15.18 If bin stores or bins are subject to misuse or are not properly maintained, preventing the regular scheduled collection of household waste then the property owner, landlord or managing agent will be required to take any necessary actions to remove waste or clear bins stores and compounds. If the Council is required to undertake any such action the costs will be charged directly to the property owner,



landlord or managing agent.

7.15.19 Any waste produced from the maintenance of the property, including construction and demolition waste, garden waste and alike, where this is produced by a contractor or service provider, or bulky items arising for disposal on change of tenancy, will not be collected by the Council as domestic waste. These are classed as commercial waste and must be collected for disposal by an appropriately registered waste contractor.

#### 7.16 Multi-occupied properties

- 7.16.1 Multi-occupied properties include licensable and non-licensable Houses of Multiple Occupation (HMOs), care homes and National Association for the Care and Rehabilitation of Offenders NACRO properties.
- 7.16.2 Householders in multi-occupied properties will be expected to make use of the full range of recycling, composting, and waste collection services as directed by the Council. These properties can apply for additional waste capacity in the same way as a normal household.
- 7.16.3 The licence holder, property owner, landlord or managing agent is required to ensure compliance with the Council's waste and recycling disposal, segregation and storage scheme as described within this policy. Waste is not allowed to accumulate within the house except where safely stored pending collection by the Council. In terms of HMO's, it is recommended that these requirements are incorporated in the HMO licence conditions.
- 7.16.4 The licence holder, property owner, landlord or managing agent shall ensure that tenants are informed of the time the waste containers must be placed outside of the property and the day of collection. This information should also be permanently displayed in a prominent position within the property.
- 7.16.5 Any waste arising from the maintenance of the property, including construction and demolition waste, garden waste where this is produced by a contractor and bulky items for disposal on change of tenancy will not be collected by the Council as household waste. This waste is classed as commercial waste, and therefore the license holder, property owner, landlord or managing agent should arrange for an appropriately registered waste collection contractor to remove this waste for disposal at an appropriate permitted facility.

#### 7.17 Places of religious worship

7.17.1 If requested, the Council will collect waste from places of worship and will not charge for collection or disposal, in accordance with paragraph 1 of Schedule 1 of



- the Controlled Waste Regulations which classifies waste from a hereditament or premises exempt from local non-domestic rating under the provisions of paragraph 11 of Schedule 5 to the Local Government Finance Act 1988.
- 7.17.2 Collections from places of religious worship shall be the same as those provided to general households and may include garden waste collections, as a chargeable service, if required. Collections will be made on the same frequency as general domestic households.
- 7.17.3 However, where the maintenance of the garden is undertaken through a gardening service; this will be classified as commercial waste, and it will be the responsibility of the gardening service to remove the garden waste.
- 7.17.4 In accordance with paragraph 11(1)(b) of Schedule 5 of the Local Government Finance Act 1988, these collections also apply to buildings used in connection with the conduct of public religious worship such as an office or church hall. However, if the religious group hires out such buildings to other persons not connected with the conduct of religious worship this is a commercial activity and the Council will make a charge for both collection and disposal in these circumstances.

#### 7.18 Premises occupied by a charity.

- 7.18.1 The Council provides 1 free collection per week and discounts for any additional collections where required but will charge for bin hire from premises occupied by a charity in accordance with Paragraph 15 of Schedule 2 of the Controlled Waste Regulations.
- 7.18.2 However, if the activities of the charity are about the conduct of public religious worship the Council will not charge for collection or disposal of collectable waste.

#### 7.19 Waste collection from community centres and village halls

- 7.19.1 Where community centres and village halls are hired out or where the use for other purposes exceeds the level of use for public meetings, this is defined as commercial waste, attracting a trade waste charge for both collection and disposal. This is enforced at the Council's discretion (see section 7.25)
- 7.19.2 A charge may be made for the collection of materials for recycling and composting in line with our Commercial Waste Service.

#### 7.20 Properties of mixed use (business and residential)

7.20.1 Mixed use properties are generally business properties with living accommodation attached e.g., a flat above a shop. Waste collections from the residential element of mixed-use properties are treated by the Council in the same manner as normal



- domestic properties. All business waste should be kept separate from any domestic waste.
- 7.20.2 The containers provided by the Council for household waste collection must not be used to dispose of business waste and any person found using containers in this way may have them removed and may be subject to prosecution by the Council under the provisions of the Environmental Protection Act 1990.
- 7.20.3 The Council require containers to be stored within the curtilage / boundary of the property. However, it is acknowledged that for mixed use properties this is not always possible. In such cases, the Council will identify and agree with the property owner, landlord or managing agent and householders a specific storage location and collection point.

#### 7.21 Residential properties also used for business

- 7.21.1 Waste produced during any activity for gain or reward, whether on business or domestic premises, while self-employed or working for others is classed as commercial waste. Businesses are legally obliged to store their waste securely, separate from domestic waste and to dispose of it responsibly using an appropriately licensed waste carrier.
- 7.21.2 Failure to comply with the duty of care requirements is a criminal offence and could lead to prosecution.
- 7.21.3 The Council will not collect through its household waste collection services waste that it believes is generated by a business at a residential property. However, the Council can, by separate arrangement, provide a commercial waste collection service for which a charge will be applied.
- 7.21.4 Where a business operates from a residential property and waste from the business is found within the household container, the container will not be collected as it will be recognised as a contaminated container and the Council may take education and / or enforcement action against the occupant that is operating the business.
- 7.21.5 If a child-minding service is operating from any dwelling, the waste generated shall be contained within the containers provided to a standard residential household. If excess waste is generated beyond the standard collection capacity provided, then this shall be treated as business / commercial waste; the collection of which will need to be arranged for separately; for which a charge for collection and disposal may apply.



#### 7.22 Collections from schools and education establishments

7.22.1 Waste from universities, colleges, independent schools, nurseries, playschools and pre- schools is classified as household waste but a charge shall be made for both non-recyclable, recycling waste and composting, including garden waste collections.

#### 7.23 Non-domestic and commercial / business waste

- 7.23.1 Section 75(7) of the Environment Protection Act 1990 defines commercial waste as 'waste from premises used wholly or mainly for the purposes of a trade or business or the purposes of sport, recreation or entertainment'.
- 7.23.2 The classification of waste is set out in Statutory Instrument No 811 the Controlled Waste (England and Wales) Regulations 2012, which came into effect on 6 April 2012. The Regulations prescribe how waste is to be treated as household, industrial or commercial waste defined by either by its source or the activity producing the waste. The regulations further define household waste for which a collection and / or disposal charge may be made.
- 7.23.3 Collections of commercial waste from whatever source will be made in accordance with the Council's Commercial Waste Services terms and conditions.

#### 7.24 Collections from outdoor events

- 7.24.1 All waste generated at outdoor events is classified as commercial waste. Event organisers and vendors have a 'Duty of Care' to ensure all waste is disposed of in a proper manner using a registered waste carrier in accordance with Section 33 of the Environmental Protection Act 1990. The Council can, by separate arrangement, provide a commercial waste collection service for which a charge will be made.
- 7.24.2 For third party organised events taking place in parks, on green spaces and in other public places the terms and conditions of hire require the event organiser to produce for approval an effective waste plan, to manage the recycling and waste management activities at the event accordingly. Persons or organisations hiring parks, green spaces or other public places will be expected to apply the principles of waste minimisation in the first instance while encouraging as much of the material as possible to be source separated for recycling or composting.
- 7.24.3 Event organisers shall be responsible for managing levels of contamination and ensuring that commercial operators [such as food vendors] fulfil their 'duty of care' responsibilities. Event organisers shall be required to ensure that vendors operating catering and refreshment concessions substitute any non-recyclable containers, cartons and containers with items made from materials that can be



recycled – for example using plastic or card that can be more readily recycled rather than polystyrene cups and trays.

# 7.25 Communication, Education and Enforcement protocol for waste and recycling services

7.25.1 The Council acknowledges the critical role of communication and education in engaging residents and businesses on waste and recycling practices. These initiatives are implemented through various channels, including recycling roadshows, direct intervention by recycling officers, targeted correspondence, and digital communication platforms such as the Basildon Council website and social media pages.

Basildon Council places a high priority on waste and recycling education, recognising its essential role in supporting our aims to enhance understanding among residents and businesses about the importance of maintaining clean and sustainable communities and protecting the local environment.

Enforcement will be undertaken in conjunction with Council's Corporate Enforcement Policy and any relevant departmental policies relating to the collection and disposal of waste.

- 7.25.2 In accordance with the provisions of the Environmental Protection Act 1990, the Council has directed householders within the Council area to use a recycling, composting and food waste collection service.
- 7.25.3 The Council considers this to be a compulsory recycling and waste collection scheme and, consequently, any persons failing to comply with these directions could be subject to education and / or enforcement action, using, but not limited to, the provisions of, the Environmental Protection Act 1990, the Clean Neighbourhoods and Environment Act 2005 and the Anti- Social Behaviour, Crime and Policing Act 2014 as appropriate.
- 7.25.4 Education and / or Enforcement action may arise from but not limited to:
  - Failure to present waste correctly and safely see section 7.3.
  - Householders disposing of unauthorised waste within provided container(s) see section 7.4
  - Householders disposing of inappropriate materials in the wrong container(s) leading to contamination – see section 7.4
  - Householders not using the container(s) for its intended purpose see section 7.2
  - Householders using inappropriate container(s) see section 7.2
  - Householders using domestic waste container(s) for business and commercial waste – see section 7.23



7.25.5 The Council reserves the right to issue a charge to the occupier under the Controlled Waste (England and Wales) Regulations 2010. The charge will be used to recover the additional collection / disposal costs and not as a means of enforcement.

#### 7.26 New developments and planning guidance

- 7.26.1 Basildon Council has made available a refuse and recycling advice note for developers, which can be found in detail here: <u>Basildon Council - Refuse and</u> <u>Recycling Advice for Developers - March 2015</u>
- 7.26.2 In summary the note assists developers and applicants by highlighting current management of refuse and recycling collections and what provisions will be excepted when proposals for new dwellings and commercial premises come forward in the future
- 7.26.3 It is important that the guidance is referred to from the earliest stages of building design. All developments should be designed with waste and recycling in mind, including adequate storage areas for waste management facilities and good access for collection crews.
- 7.26.4 It is advised that all applications for new developments are to be accompanied by a concise waste management report that addresses the relevant aspects of the guidance notes provided by the Council in the advice for developers.

# 8 What evidence have you used that suggests your Policy approach is the right one?

The policy has been drafted following a review of the Council's existing policies and procedures related to waste and recycling as well following research on other authorities related policies.

Our approach is based on best practice, meeting the changing waste and recycling environment and the objectives of the Council. Alongside this Equalities Impact Assessment (EIA) has been completed to ensure changes to Council practices and procedures, as in impact of this policy does not detrimentally impact residents.

# 9 Who or what will be affected by the Policy?

All borough residents, businesses and customers of the waste and recycling services provided by the Council, and any other party involved in the enforcement of this policy.



#### 9.1 Inclusion and Diversity

An Equalities Impact Assessment has been completed, which has led to adjusting or adapting the policy to reduce negative impacts such as:

- Reasonable adjustments to containers and service provisions, such as different size or types of containers to be used for the collection of waste by the Council, plus changes to the presentation of the Council's waste policies.
- Increased promotion of assisted collections
- Development of eligibility criteria for additional non-recyclable waste capacity
- Availability of materials in differing languages
- Utilisation of all communication channels and clear and concise communication material

# 10 Corporate Knowledge

	Levels of Impact			
Corporate Ambitions	High	Medium	Low	None
We want Basildon to be home to			<u>X</u>	
healthy and active local				
communities able to support				
themselves and each other.				
We want Basildon to offer a high	<u>x</u>			
quality of life for all				
householders through attractive,				
liveable, accessible and safe				
neighbourhoods and towns				
along with the provision of				
enduring facilities, green spaces				
and town centres that meet the				
needs of the community.				
We want Basildon to have a			<u>x</u>	
thriving, dynamic and diverse				
economy where all our				
communities benefit from				
increased opportunity and our				
workforce has the right skills for				
our local economy and beyond.				



## 11 Accompanying Strategy

Waste Strategy (2020-25),

#### 12 Links to other Corporate Policies or Partner documents

- Corporate Plan
- Local Plan
- Climate Change Policy
- Climate Change Policy (2021-24)
- Environment Act (2021)
- Government Resources and Waste Strategy (2018)
- Waste Strategy for Essex 2024 2055

# 13 Appendices

#### 13.1 Glossary of terms and conditions

Assisted Collection: A service for eligible residents to support the collection of household waste from an agreed collection point on a person's property due to difficulty of moving container(s) to the boundary of their property.

Bulky Waste: A charged on-demand waste collection service for domestic properties disposing of large items which do not fit into container(s) or exceed safe weight limits.

Clinical waste: Waste including healthcare and clinical waste, such as syringes, needles, human or animal tissue, swabs or dressings, excretions, and drug and pharmaceutical products from domestic products using appropriate containers such as sharp bins or clinical waste sacks.

Collection Point: An agreed location for the placement of containers for the collection of materials by the collection crews. These are generally at the edge of the property boundary but may be elsewhere as agreed by the resident and the Council.

Commercial Waste: Waste from premises used wholly or mainly for the purposes of a trade or business or the purposes of sport, recreation, or entertainment'

Contamination: When unauthorised waste is included in the wrong container(s), for example food waste found in dry recycling collections, these will not be collected by the Council.



Dry Recyclable materials: Clean recyclable materials including paper and card, metal and cans, plastics, tubs, and glass.

Education: The act to inform residents of the service and how to dispose of their waste correctly

Enforcement: The act to make residents follow law, policy and expectations through actions including education and fines for example.

Excess waste: Waste that is outside of the required container(s) that is not able to be collected.

Food waste: Organic waste from all food related items, excluding liquids.

Garden Waste: Organic waste from gardens, including grass cuttings, leaves and bushes etc.

Hazardous Waste: Any type of waste that when present in quantities and concentrations that are high enough, to pose a threat to human health or the environment if they are improperly stored, transported, treated or disposed. Examples include batteries, pesticides and chemicals.

Household / Domestic waste: Any waste that is generated by households, excluding those from commercial works.

Houses in Multiple Occupation (HMO): An HMO is a house or flat that is let to 3 or more unrelated people (or 2 or more households) and who share basic amenities, i.e., a kitchen, bathroom, or toilet. All HMOs within Basildon Borough must be registered with Basildon Council.

Incontinence / Offensive / Hygiene Waste: Is non-infectious waste, which is unpleasant and may cause offence to those encountering, for example nappies, incontinence pads, animal waste etc.

Missed Collection: Any container(s) that has not been collected by crews due to no other reason than missing the collection.

Non-Standard kerbside collection properties: These properties have adjustments to the current kerbside collection model for example, flats, communal collection points and



hard to access areas. They are likely to have different collection containers, or different ways of sorting waste materials.

Residual waste: All non-hazardous, non-recyclable and non-clinical waste

Side Waste: additional waste items placed out for collection not presented in the correct containers provided by the Council.

Standard kerbside collection properties: The majority of properties that have a standard collection service with no adjustments.

Unauthorised waste: Waste that the Council do not collect, or do not collect within that collection service.



# **Translations**

If a translator is required, call 01268 208274 - press \* when messages start playing or select an option number from the languages below:

(Romanian) Sunați la 01268 208274 și selectați opțiunea 1 pentru limba română

(Polish) Proszę zadzwonić pod numer 01268 208274 i wybrać opcję 2 dla języka

polskiego.

(Lithuanian) Surinkite 01268 208274 ir norėdami pasirinkti lietuvių kalbą spauskite 3

(Malayalem) മലയാളത്തിനായി നാല് അമർത്തൂ

(Portuguese) Ligue 01268 208274 e prima a opção 5 para português

للغة العربية اتصل على الرقم 208274 01268 أضغط الخيار رقم 6 (Arabic)

با شماره تلفن208274 01268 تماس بگیرید وبرای زبان فارسی، 7 را فشار دهید

(Tigrigna) ብ01268 208274ብምድዋልንት ግርኛ ኣጣራ ኢ8 ጠው ቅ

تەلمەفۇن بكە بۆ ژمارە\$208274 01268 و بۆ زمانى سۆرانى بژاردەى 9 داگرە

با شماره تلیفون208274 01268 به تماس شوید و برای لسان دری، گزینه 10 را فشار دهید

د208274 01268شمېرې ته زنګ وو هئ او د پښتو لپاره11کښېکاږئ (Pashto)

(Ukrainian) Зателефонуйте за номером 01268 208274 і натисніть опцію 12 для

переходу на українську мову

For anything else please press \*